Author's response to reviews

Title: Melasma and its association with different types of nevi in women: A case control study

Authors:

Hassan Adalatkhah (h.adalatkhah@yahoo.com)
Homayoun Sadeghi-Bazargani (homsad@ki.se)
Nayereh Aminisani (n.aminisanibeja@yahoo.com)
Zeynizade Somayye (Somayye.zeynizade@yahoo.com)

Version: 2 Date: 14 January 2008

Author's response to reviews:

As requested abstract format is corrected.
reference format changed using reference manager
This manuscript lacks a figure but I couldn't change the 1 figure information at manuscript details page.